Author's response to reviews

Title: Genetic polymorphisms involved in dopaminergic neurotransmission and risk for Parkinson's disease in a Japanese population

Authors:

Chikako Kiyohara (chikako@phealth.med.kyushu-u.ac.jp)
Yoshihiro Miyake (miyake-y@fukuoka-u.ac.jp)
Midori Koyanagi (koyanagi@fukuoka-u.ac.jp)
Takahiro Fujimoto (tfujimoto@fukuoka-u.ac.jp)
Senji Shirasawa (sshirasa@fukuoka-u.ac.jp)
Keiko Tanaka (k-tanaka@fukuoka-u.ac.jp)
Wakaba Fukushima (wakaba@med.osaka-cu.ac.jp)
Satoshi Sasaki (sstssasak@m.u-tokyo.ac.jp)
Yoshio Tsuoboi (tsuoboi@cis.fukuoka-u.ac.jp)
Tatsuo Yamada (tyamada@fukuoka-u.ac.jp)
Tomoko Oeda (toeda@unh.hosp.go.jp)
Hiroyuki Shimada (h.shimada@med.osaka-cu.ac.jp)
Nobutoshi Kawamura (nobutosi@neuro.med.kyushu-u.ac.jp)
Nobutaka Sakae (sakae@neuro.med.kyushu-u.ac.jp)
Hidenao Fukuyama (fukuyama@kuhp.kyoto-u.ac.jp)
Yoshio Hirota (hiro8yoshi@med.osaka-cu.ac.jp)
Masaki Nagai (mnagai@saitama-med.ac.jp)
Fukuoka Kinki Parkinson's Disease Study Group (chikako@phealth.med.kyushu-u.ac.jp)

Version: 3 Date: 21 July 2011

Author's response to reviews: see over
Dear Sir,

We are returning herewith our manuscript revised according to your email of July 20, 2011. As suggested, we have made major and minor revisions. Again, thank you very much.

Sincerely yours,

Chikako Kiyohara, Ph.D.
Department of Preventive Medicine
Kyushu University
Graduate School of Medical Sciences
Assistant Professor